 Vc/13-May-04
Week20 starting 10 May 04

SM18 Operation Meeting (Tuesdays)

 Bench status  and tests for now & next few days : 

	Bench
	Name
	STATUS 10/05 afternoon
	 

	TBA1
	_
	Not Activated
	 

	TBA2
	3117 ?
	Commissioning & RETEST
	This week ?

	TBB1
	MBAR2039
	Prep 1 Lyre Test
	300 K

	TBB2
	MBAR3024
	PT 12 Warm Up
	Warming

	TBC1
	MBAL3148
	PT 12 Warm Up
	Warming

	TBC2
	SSS015
	PT 12 Warm Up
	Warming

	TBD1
	MBBL2061
	Prep 5 Pumping + Cool Down
	Cooling

	TBD2
	MBBL2078
	OUT OF USAGE
	300 K

	TBE1
	MBBR3016
	PT 1 IAP @ cold
	1.9 K

	TBE2
	MBBL1114
	PT 12 Warm Up
	Warming

	TBF1
	MBBL3135
	CR 1 Leak Test
	300 K

	TBF2
	MBBL1097
	ICS 5 Traveller Last
	 

	 
	 
	 
	 


Test Coordinator this week: Marco ; MTM support on call  Giorgio:  (16-3602).

10 Dipoles  Disconnected after tests last week    ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Repair of D2 Needs All benches above 80’ K

· 36 hrs required ?    Plan warm up evening so work starts morning ?

A2  commissioning & re-test 3117 ?

· General Problems last week :  Cable rupture on F1  for Anticryostats
·  SSS work :   GREY FILE IN SMCR  ….   To do list etc Nair + Sridhar , 
	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday 04_N
	TBD1
	No possibility to close or open the electro-valves on the benches


	MTM
	
	Guy came and realised that the  PLC valves control was not working. He opened the valves manually.

Origine: Electrical overcharge which burned the fuse. Now the system is working (Federico)

	Tuesday 04_N
	TBD1
	2049 magnet: oscillation present during the discharge of HF heaters at 1500 A.
	MTM
	
	3 kA min energy for diode test performed: diagnostic confirmed. Test was stopped.

Magnet was warmed up Wednesday afternoon because the QH YT211 is not working.

	Wednesday 05_M
	TBC1
	IAP@warm:

1) HWREC problem which was solved.

2) TEMA did not load the proper configuration
	MTM
	HWREC+TEMA
	Many attempt. Still not OK. Reset of the racks then OK.

	Wednesday 05_M
	TBC1
	HV test stops. Launch 3 times run-recovery without success.
	MTM
	
	Rack changed.

	Wednesday 05_A
	ALL
	SMTMS temperatures were not refreshed.
	ACR
	
	J.Axensalva working on the problem

	Wednesday 05_A
	TBD1
	2049 HV test @cold: Lost of the cryo OK .
	ACR
	
	

	Wednesday 05_N
	TBF1
	HF config. File not found. No data was found during the second training quench
	AB-CO
	HF
	

	Tuesday 06 _M
	TBE1
	3016: Mobic did not accept the scanned data.
	MTM?
	HWREC
	Wrong data in MTF for the anti-cryostats.

	Tuesday 06 _M
	?
	IAP@warm not possible: no trigger.


	AB-CO
	TEMA
	TEMA re-started . Then OK

	Tuesday 06 _A
	TBF1
	Magnetic measurements: Wrong power supply selected (TBB). 
	MTM
	Power
	Power supply changed manually.

	Tuesday 06 _A
	TBE1
	HV test at warm on 3016. Problem with the first sequence.
	MTM
	HV
	Richard informed. He said that the capacitance values reach the limits. Test is OK

	Tuesday 06 _N
	TBF1
	Full load line stopped because power supply went off.
	AB-PO??
	
	Cycle re-started.

	Tuesday 06 _N
	TBC2
	Power supply could not be switched on.
	MTM?
	
	Guy came and found the digital communication cable was disconnected on PC.

He also reboots the sun MTA19.

	Tuesday 06 _N
	TBC1
	HRWEC problem: INTERCOM missing.
	?


	HWREC
	Guy created the files manually. HWREC off for this test.


	8Sl. No
	Date /Day
	Shift
	Bench/Cluster
	Magnet
	Problems
	Remarks 

	1
	8/5/04- Sat
	N
	D1
	MBBL 2061
	TBD1: leak problem 1*10-3 on the manchette. ICS will try to repar tomorrow.
	

	2
	8/5/04- Sat
	N
	F1
	MBBL 3135
	TBF1 : The long anti cryostat cable was broken, ICS missing to disconnect this cable, when rocla came the wire was broken.
	

	3
	8/5/04- Sat
	N
	Rocla Problm
	---
	Rocla problem (bolding) a responsable (Pascal) is comming, the 1st rocla have a bolding problem, the 2nd will be repared for its electrical problem, monday morning.
	

	4
	8/5/04- Sat
	N
	E2
	MBBL

1114
	TBE2 : Always a temperature sensor problem, a 0-200mA power supply has been taken in work shop. But not possible to have cryo OK.
	

	5.
	8/5/04- Sat
	M
	F2
	MBBL

1097
	TBF2: Anticryostat alarm after breaking the vacuum - temperature dropped to 0C. Temp. started rising after a while.
	PLC reset was done. 

	6
	8/5/04- Sat
	M
	E2 
	MBBL

1114
	Cryo temperature sensor (TT 130B) on one of the current leads not working.  So cryo O.K. signal lost.
	To be solved by cryo equipment support

	7
	7-8/5/04 Fri-Sat
	N/M
	F2
	MBBL

1097
	Safety Matrix Rack on which Min Energy card is mounted is receiving permanent triggers. So 3KA diode test could not be carried out.    The Min NRG card was disabled by morning shift. The diode test can be carried out in the afternoon shift


	No triggers are seen when bench F1 is selected.

	8
	7/05/04

Fri
	N
	E2
	MBBL

1114
	 Problem of temperature sensor on the CFB, the current lead are freeze.


	

	9
	7/05/04

Fri
	N
	F2
	MBBL

1097
	TBF2: Equipment Support had been called for a trigger always present on matrix Min. NRJ card, couldn't be solved, we continue the training because we can do this Diode Test tomorow.
	

	10
	7/05/04

Fri
	N
	E2
	MBBL

1114
	TBE2: The problem will be resolved tomorow, because it is not possible to pass on an auxiliary sensor temperature, to give the cryo ok 1.9K.
	

	11
	7/05/04

Fri
	N
	F2
	MBBL

1097
	TBF2: first training quench 2 files genereted (.a01 only for HF), full data in .a02
	

	12
	6/05/04

Thu
	N
	F1
	MBBL 1043
	TBF1: Full load line stopped and power supply off so we had to restart again
	

	13
	6/05/04

Thu
	N
	C2
	SSS
	TBC2:problem for switching on the power supply so we called Guy.
	

	14
	6/05/04

Thu
	N
	C2
	SSS
	TBC2: digital communication cable was disconnected on PC. Reconnected then OK
	

	15
	6/05/04

Thu
	N
	C2
	SSS
	TBC2: guy reboot sunmta19
	








